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Device Service Record Form for TB9AES028
	Date
	Comments

(Include the following in each MSR entry: What, Where, Why)
	Info Provided By

	3/25/14
	Received in IB4
	JB

	4/8/14
	Leak Test and Visual Inspection completed in IB4 
	DP

	4/17/14
	In IB4 RF clean room for income RF QC
	M. Hassan

	4/17/14
	RF QC is complete. F=1299.307 MHz Q=9320 FF=57% Ecc=0.4 mm.
	M. Hassan

	4/17/14
	Cavity is back to crate
	M. Hassan

	4/22/14
	CMM inspection complete.
	R. Riley

	4/23/14
	Cavity transported from IB4 to ICB for optical inspection.
	Jim Rife

	4/25/14
	Cavity moved to ICB OPT INSP.
	Jim Rife

	4/28/14
	Optical Inspection is complete.  Cavity transported to ICB storage.
	Jim Rife

	5/1/14
	Cavity transported from ICB to IB4.
	Jim Rife

	5/7/14
	Cavity moved to ICB for storage
	JB

	5-16-2014
	Cavity transported from ICB to ANL
	Rick Espinoza

	5/16/14
	Cavity received at ANL
	Chris Baker

	6/11/2014 – 6/12/2014
	Cavity received 170µm EP over the course of 2 days.
	Ryan Murphy

	6/13/14
	Cavity transported from ANL to MP9.
	Jim Rife

	6/16/2014
	Cavity baked w N2 in MP9
	M. Merio

	6/18/14
	Cavity transported from MP9 to ANL.
	Jim Rife

	6/18/14
	RF tuning Completed, Freq=1298.2 MHz  FF=98% Ecc=0.55 mm. Cavity located in IB4 furnace area.
	PB

	06/19/14
	Cavity received at ANL
	Chris Baker

	6/24/2014
	Cavity received 5µm EP at ANL
	Ryan Murphy

	7/8/14
	Cavity received HPR/assembly sequence at ANL
	Chris Baker

	7-9-2014
	Cavity transported from ANL to ICB
	Rick Espinoza

	7-14-2014
	Cavity moved from ICB to IB1
	Rick Espinoza

	7/17/14
	Cavity transported from IB1 to IB4.
	Jim Rife

	7/17/14
	Cavity disassembled in IB4 cleanroom
	Chad Thompson

	7/17/14
	Cavity transported from IB4 cleanroom to ICB OPT INSP.
	Jim Rife

	7/18/14
	Optical Inspection complete.
	Jim Rife

	7/18/14
	Cavity transported from ICB optical inspection to IB4 CPL
	Chad Thompson

	7-18-14
	Prepped cavity for tumbling with ACT M need to remove 125microns loaded cavity into tumbler
	Dburk

	7-19-14
	Started tumbling
	Dburk

	7-19-14
	Cleaned out cavity with low pressure rinse. While rinsing the LPR malfunctioned and the wand stopped working halfway into cavity tried to remove the wand by retracting the wand in the horizontal position. The wand did not retract but it did go forward. The wand sagged and struck just below the 4th iris. Inspected the cavity and seen a ding where the sprayer head hit the cavity. A.C. was notified and was decided that we needed to tumble with ACT M again to remove ding and to remove more material to achieve our goal of 125microns (7.5hours removed about 70microns) the LPR was repaired and the problem was a wire that was pulled out of a connector causing the LPR to run erratically.  
	Dburk

	7-22-14
	Prepped for tumbling ACT M 
	Dburk

	7-23-14
	Started tumbling
	dburk

	7-23-14
	Stopped tumbling after 7 hours cleaned out and prepped for cones
	Dburk

	7-24-14
	Started tumbling with cones stopped after 13 hours cleaned out and prepped for 600 mesh and wood blocks
	Dburk

	7-25-14
	Started tumbling with 600 mesh wood blocks
	Dburk

	7-28-14
	Stopped tumbling after 37.5 hours cleaned 
	Dburk

	7/29/14
	Cavity moved from IB4 CPL to ICB OPT INSP
	Chad Thompson

	7/30/14
	Optical inspection complete.  Cavity moved from ICB OPT INSP to IB4 cleanroom in preparation for HPR
	Chad Thompson

	8-4-2014
	Cavity transported from IB4 to ANL
	Rick Espinoza

	08/04/14
	Cavity received at ANL
	Chris Baker

	8/6/2014
	Cavity received 60µm EP at ANL
	Ryan Murphy

	8-11-2014
	Cavity transported from ANL to MP9 for furnace treatment
	Rick Espinoza

	8/11/2014
	Cavity baked 2 N2 in MP9
	M. Merio

	8/14/2014
	Cavity in IB4 RF Lab for tuning. QC is done, F=1296.21 MHz, Q=9261, Ecc=0.67 mm, FF=73%
	M. Hassan

	8/14/2014
	Tuning is complete. , F=1296.51 MHz, Q=9035, Ecc=0.67 mm, FF=99%
	M. Hassan

	8/18/14
	Cavity transported from IB4 to ANL.
	Jim Rife

	8/18/14
	Cavity received at ANL
	Chris Baker

	8/19/2014
	Cavity received 5µm EP at ANL
	Ryan Murphy

	8/22/14
	Cavity received HPR/assembly sequence at ANL
	Chris Baker

	8/25/14
	Cavity transported from ANL to IB1.
	Jim Rife

	8/26/14
	VTS test results from T&I e-log.  57260
	Jim Rife

	8/26/14
	VTS test results from T&I e-log.   57263
	Jim Rife

	8-29-2014
	Cavity moved from IB1 to IB4 RF lab
	Rick Espinoza

	10/27/2014
	Cavity has been tuned. F=1297.9 MHZ Q=9070 Ecc=0.42 mm FF=99% L=1010.47 mm
	M. Hassan

	10-31-2014
	Cavity transported from IB4 to ANL
	Rick Espinoza

	10/31/14
	Cavity received at ANL
	Chris Baker

	11/5/14
	Cavity received HPR/helium weld prep assembly at ANL
	Chris Baker

	11-6-2014
	Cavity transported from ANL to MP9
	Rick Espinoza

	11/19/14
	Cavity going to IB4 (picked up by Rick Espinoza)
	Mark Chlebek

	11/24/14
	Cavity transported from IB4 to MP9 
	Bgolden 

	11/24/14
	RF tuning completed, F=1297.89, FF=99% Ecc=0.49 mm, Cavity located in IB4 furnace area.
	PB

	12/8/14
	Cavity at IB4 CPL cleanroom, prepped for weld at MP9
	Chad Thompson

	2/13/15
	Cavity location verification.  Cavity at MP9
	Jim Rife

	3/06/15
	Cavity transported from MP9 to ANL for VTS prep 
	Bgolden

	3/6/15
	Cavity received at ANL
	Chris Baker

	3/8/15
	Cavity received HPR/assembly sequence at ANL
	Chris Baker

	3-9-2015
	Cavity Transported from ANL to ICB for holding until IB1 is ready to receive it for VTS
	Rick Espinoza

	3/17/15
	Cavity transported from ICB to IB1 for testing.
	Jim Rife

	4/3/15
	Test results from T&I e-log.   60403
	Jim Rife

	4/10/15
	Cavity transported from IB1 to IB4, Endgroup leakcheck, FF measurement 
	Bgolden 

	4/16/15
	Cavity transported from IB4 to ICB; awaiting next step helium vessel hardware prep
	Bgolden

	4/17/2015
	Please, notice that cavity still needs final RF tuning prior to vessel welding
	M. Hassan

	4/24/15
	Cavity moved from ICB to RF lab for RF QC
	Damon Bice

	4/29/15
	RF QC is completed. F=1297.84 MHz Q=9086 FF=98.09% Ecc=0.44 mm L=1008.49 mm.
	M. Hassan

	5/07/15
	RF QC before FF final tuning is completed. F=1297.83 MHz Q=9239 FF=98.3% Ecc=0.45 mm L=1009.9 mm.
	A. Lunin

	5/08/15
	RF complete tuning (before HV welding) is completed. F=1297.94 MHz Q=9075 FF=98.74% Ecc=0.39 mm L=1010.3 mm.
	A. Lunin

	5/27/15
	RF QC check. F=1297.93 MHz Q=9082 FF=98.4% Ecc=0.36 mm L=1009.59. Cavity is ready to proceed for HV welding.
	M. Hassan

	6/1/15
	Cavity moved to IB4 cleanroom for HV weld prep
	Chad Thompson

	6/17/15
	Location update --- Cavity currently in MP9
	Damon Bice

	6/23/15
	Cavity at ICB main floor
	Mark Chlebek

	6/25/15
	Cavity transported from ICB to ANL.
	Jim Rife

	6/25/15
	Cavity received at ANL
	Chris Baker

	7/2/15
	Cavity received HPR/assembly sequence at ANL
	Chris Baker

	7/7/15
	Cavity transported from ANL to ICB awaiting testing.
	Jim Rife

	7/14/15
	Cavity transported from ICB to IB1 for testing.
	Jim Rife

	7/16/15
	Cavity test results from T&I e-log.    62392
	Jim Rife

	7/21/15
	Cavity transported from IB1 to ANL.
	Jim Rife

	07/21/15
	Cavity received at ANL
	Chris Baker

	07/26/15
	Cavity received HPR/Assembly sequence at ANL
	Chris Baker

	7/27/15
	Cavity transported from ANL to MP9.
	Jim Rife

	8-11-2015
	Cavity received at Meson for HTS qualification testing.
	Rick Espinoza

	9/11/15
	Cavity transported from MP9 to ICB for Optical Inspection.
	Jim Rife

	9/15/15
	The non-recorded Optical Inspection is complete.  The Cavity was transported to ANL for processing.
	Jim Rife

	09/15/15
	Cavity received at ANL
	Chris Baker

	09/17/15
	Cavity received 2um EP at ANL
	Chris Baker

	09/23/15
	Cavity received HPR/assembly sequence at ANL
	Chris Baker

	9/24/15
	Cavity transported from ANL to ICB.
	Jim Rife

	9/28/15
	Cavity transported from ICB to IB1 VTS.
	Jim Rife

	10/7/15
	Cavity transported from IB1 to MP9.
	Jim Rife
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